\documentstyle{article}
\begin{document}
\section{Instructions to install MOSAICO}
Instructions to install MOSAICO from a tape (TK50):

{\tt cd} to any directory, for example {\tt \~ octtools/src}

{\tt tar x}

You need about 5M for the MOSAICO sources and about 20M for the
examples provided.

You will see two new directories: {\tt./MOSAICO , ./examples}

{\tt cd MOSAICO }

{\tt make install >\& make.out }

or just {\tt make}

If everything is OK you can clean up with {\tt make clean}.

All the tools are installed in {\tt \~ octtools/bin} and the man pages in
{\tt \~ octtools/man/man1  \~ octtools/man/man3}.

Also something gets installed in {\tt \~ octtools/lib/technology/scmos}.

Make sure this directory is write accessible.

Some examples are provided in {\tt ./examples}. For example see 
{\tt ck}. 

{\tt cd ./examples/ck}

{\tt mosaico ck:puppy |\& tee mout}


\end{document}
